Energy price shocks, conflict inflation, and income distribution in a three-sector model

Rafael Wildauer, Karsten Kohler, Adam Aboobaker, Alexander Guschanski





Please cite this article as: R. Wildauer, K. Kohler, A. Aboobaker et al., Energy price shocks, conflict inflation, and income distribution in a three-sector model. *Energy Economics* (2023), doi: https://doi.org/10.1016/j.eneco.2023.106982.

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

© 2023 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

Energy Price Shocks, Conflict Inflation, and Income Distribution in a Three-sector Model^{*}

Rafael Wildauer[†]

Karsten Kohler[‡]

ıler[‡] Adam Aboobaker[§]

Alexander Guschanski[¶]

July 2023

Abstract

The paper presents a model of conflict inflation to investigate the distributional effects of energy price shocks. We argue that periods of high inflation are always periods of significant redistribution of income. We analyse how such redistribution occurs along two dimensions: between workers and firms and between sectors of the economy. To study the distributional outcomes of the recent inflationary episode, we build a three-sector model comprising a domestic energy sector which provides inputs for a goods and a services sector. The model is calibrated to US sectoral data with the Method of Simulated Moments. While energy prices are set internationally, non-energy prices and nominal wages are set by firms and workers, giving rise to conflicting claims over the distribution of income. We consider three shocks that trigger inflationary distributional conflict: an energy price shock combined with demand and supply shocks to the goods sector. We find that the recent inflationary episode constitutes a price-wage rather than a wage-price spiral. The combined shocks induce non-energy firms to raise prices, which undermines real wages, and redistributes income towards firms. The sectoral demand shift towards goods in combination with pandemic-related supply bottlenecks further raises mark-ups, accelerating inflation and leading to divergence in sectoral profit margins. We compare three anti-inflationary policies: redistributing windfall profits to workers, nominal wage restraint, and aggregate demand contraction through monetary or fiscal policy. The redistribution of profits via a windfall tax is most effective in reducing inflation without reinforcing reductions in employment and labour shares.

Keywords: energy price shocks, inflation, income distribution, multi-sector model, wageprice spiral, price-wage spiral

JEL codes: E24, E31, J30

^{*}We thank Esra Ugurlu, Peter Skott, and participants at an IPE Berlin research seminar for helpful comments. Rafael Wildauer and Alexander Guschanski thank the Hans Böckler Foundation for financial support through grant number 2021-544-2. Any errors are our own.

[†]University of Greenwich, Greenwich Business School. E-Mail: r.wildauer@gre.ac.uk

[‡]Economics Department, Leeds University Business School. E-Mail: k.kohler@leeds.ac.uk

[§]University of the Witwatersrand, School of Economics and Finance and SOAS University of London, Department of Economics. E-Mail: adam.aboobaker@wits.ac.za

[¶]University of Greenwich, Greenwich Business School. Email: Alexander.Guschanski@gre.ac.uk

1 Introduction

After more than a decade of sluggish price growth in advanced economies, inflation is back. Since the summer of 2021, many advanced countries have seen inflation rates above ten percent - well beyond most central bank inflation targets. There appears to be broad agreement that key drivers of the recent inflation surge include supply-chain disruptions and shifts in consumer demand caused by the COVID-19 pandemic, amplified by increasing energy prices which peaked as a result of Russia's occupation of Ukraine (Bernanke & Blanchard 2023, Bräuning et al. 2022, Kilian & Zhou 2022*a*, Storm 2022, Shapiro 2022, Weber et al. 2022).¹

While the macroeconomic policy debate is strongly focused on how to bring inflation rates down, there is a growing empirical literature documenting the distributional effects. A striking phenomenon is that adverse shocks were accompanied by a rise in profit margins in several sectors. Recent evidence documents record profits of oil firms like Shell and BP.² Notably, increases in profit margins are not limited to energy producers. Hayes & Jung (2022), Weber & Wasner (2023) and Unite (2022) show that some firms increased their profit margins considerably between 2019 and 2021. Konczal & Lusiani (2022) find that firms that enjoyed considerable mark-ups before the pandemic further raised them since the lifting of COVID restrictions, demonstrating an association between pricing power and inflation. In addition, several papers argue that firms in the goods sector increased mark-ups because they struggled to keep up with rampant demand due to a shift in consumption from services towards goods since the pandemic (Jorda et al. 2020, Furman 2021, Storm 2022). This change in the composition of demand is well-documented in the U.S., where the fraction of consumer spending on goods increased by about 13% between 2020 and 2021 (Figure 1, panel a), while expenditure on services declined by roughly 7% (Storm 2022). This shift together with disrupted supply chains and shortages in crucial inputs like microchips left many companies struggling to increase their production levels and but allowed them to increase their markups. The resulting increase in profit shares highlights the fact that periods of high inflation do not only generate losers, but also winners (Figure 1, panel b).

Crucially, increases in mark-ups do not occur uniformly across the economy, instead the recent empirical evidence demonstrates a high degree of heterogeneity across firms and sectors (Hayes & Jung 2022, Konczal & Lusiani 2022, Storm 2022). Thus, aggregate inflation coincides with changes in both sector-specific profitability and in real wages. In other words, periods of high inflation are periods of income redistribution. However, what mechanisms determine which groups and sectors benefit or lose is an open question which has received relatively little theoretical attention.

This paper develops a three-sector model of inflation dynamics, building on the underutilised theory of conflict inflation in which conflicting claims over the distribution of income deter-

¹The role of fiscal policy amidst pandemic-driven constraints to the supply-side has also been debated; see Baqaee & Farhi (2022), di Giovanni et al. (2022), Jorda & Nechio (2022) and Shapiro (2022).

 $^{{}^{2}\}mathrm{BP}$ reported record profits of \$ 30 billion in the first three quarters of 2022 compared to \$ 14.9 billion in the first three quarters of 2019.



Figure 1: US corporate profit shares and consumer demand for goods

The share of goods in private consumption expenditure (PCE) in panel (a) is based on BEA Table 2.3.5 and profit shares in panel (b) are defined as corporate profits before tax (without IVA and CCAdj) for domestic industries (BEA Table 6.17D) over final output (value added). For sector definitions see Appendix F.

mine the growth rate of nominal wages and prices (Rowthorn 1977, Blanchard 1986, Blecker & Setterfield 2019, Ratner & Sim 2022, Hein 2023, Lorenzoni & Werning 2023). By extending the conflict inflation approach to a three-sector economy, we make three contributions to the literature: First, our multi-sector framework allows us to link the defining features of the recent macroeconomic environment: spiking energy prices, shifting consumer spending habits and supply bottlenecks combined with high corporate profitability. We thus go beyond approaches that only focus on one of these aspects in isolation (Guerrieri et al. 2021, Ratner & Sim 2022) or use a one-sector framework, which cannot explain profit heterogeneity (Setterfield 2022, Lorenzoni & Werning 2023). Second, we empirically calibrate the structural parameters of our model with the Method of Simulated Moments using industry-level time-series data for the US (Franke & Westerhoff 2012, Franke 2022, Reissl 2020). Third, we use the model to analyse and compare different anti-inflationary policies with respect to their impact on inflation and income distribution.

Our analysis yields four main results. First, a pure energy price shock leads to higher aggregate inflation and a reduction in profit *margins* in non-energy sectors, while energy profit margins increase. Non-energy sector profit *shares* may increase, depending on the bargaining power of firms relative to workers. Thus an increase in energy prices alone can explain rising profit shares but not the increases in profit margins outside the energy sector. Second, a combination of higher energy prices, a shift in consumer demand towards goods, and supply bottlenecks in the goods sector lead to higher inflation as well as higher profit margins in the energy and goods sector. Third, distributional outcomes in response to these shocks are fought out based on relative pricing power between firms in different industries and between firms and workers. Thus distributional conflict arises along multiple dimensions. Fourth, our policy analysis shows

that a wide range of measures is effective in reducing inflation (windfall taxes, contractionary aggregate demand policy as well as wage restraint); however, out of these only a windfall profit tax combined with household transfers succeeds in curbing inflation while stabilising real incomes and employment.

By providing a theoretically rigorous account of the recent surge in inflation, our paper offers clarity to an ongoing public debate about wage-price spirals versus 'profiteering' (or 'price gouging'). In our model higher inflation rates are triggered by two factors. Firstly, by higher energy prices which result in increased profit margins in the energy sector. Secondly, by increased markups in the non-energy sectors which are in turn caused by the shift in consumer spending towards goods and supply bottlenecks. Subsequently, workers experiencing a fall in their living standards raise nominal wages to defend their real wage. The increase in nominal wages raises firms' costs, which are partially passed on to prices, and thereby amplify the inflationary shock. We call the described process a *price-wage spiral*. Our model allows us to formally distinguish the former from a *waqe-price spiral* which is triggered by an increase in workers' nominal wage demands. In contrast a price-wage spiral is triggered by an increase in firms' input costs or target profit margins. While both spirals lead to higher inflation rates, wage-price spirals lead to an increase in real wages whereas price-wage spirals lead to falling real wages as well as rising profit margins within the sector which triggered it.³ Highlighting the adverse distributional outcomes of price-wage spirals is a key objective this paper shares with the proponents of 'profiteering'. However, in our framework markups can only increase in response to excess demand (due to a sector-specific demand shift or supply bottlenecks); thus, demand and supply shocks are a key component of our argument.⁴ Overall, we argue that the conflict inflation framework in this paper reconciles the different aspects highlighted in the debate and helps identify policies such as re-distributive taxes that both mitigate inflation and improve the distribution of income.

The rest of the paper is organised as follows: Section 2 presents the model, section 3 discusses the empirical calibration of the structural parameters, section 4 presents simulation results and considers three different policy scenarios. Section 5 concludes.

2 Model

Our model builds on the existing literature on conflict inflation, which has mostly been developed in the post-Keynesian tradition (Rowthorn 1977) and only recently integrated into New Keynesian frameworks (Lorenzoni & Werning 2023). Specifically, our model extends the approach of Setterfield (2022) and Setterfield (2006), and is closely related to Hein & Stockhammer (2010), Argitis & Dafermos (2011), and Martins & Skott (2021) as well as the conceptual argument in Weber & Wasner (2023). Our approach however differs in several important aspects. First,

 $^{{}^{3}}$ IMF (2022) has a more loosely defined notion of a wage-price spiral as a period where both wage- and price inflation temporarily accelerates. Such a definition is consistent with our model but does not allow for a distinction between wage-price and price-wage spirals.

⁴Some accounts have suggested that excess demand might not be necessary if the energy cost shocks lead to a decline in the price elasticity of demand (Weber & Wasner 2023, Donovan 2023).

we develop a framework which takes the profits generated in domestic energy sectors explicitly into account, instead of modelling the inputs they produce as entirely imported (e.g. Hein 2014, chap. 5.2). This allows us to analyse the effect of energy price shifts in countries with a domestic energy sector. Specifically, we can distinguish the effect of energy price shifts on the distribution of income between capital and labour (captured by the profit share) and on the profitability between sectors (captured by profit margins).⁵ Second, the combination of a shift in the composition of demand, supply bottlenecks in the goods sector, and the existence of a capacity utilisation-elastic mark-up (in line with, e.g., Kaldor 1985 and Flaschel & Skott 2006) allows us to replicate the observed divergence in profit margins across sectors (Weber & Wasner 2023).⁶ Third, workers in our model target a real wage denominated by the final good price rather than a sector-specific wage share, which makes the modelling of wage demands more realistic.

We start with a two sector model with a domestic energy sector that provides intermediate inputs to a domestic final output sector. This two sector framework allows us to develop some of our main results in an analytically tractable way. We will show that a domestic energy sector plays a significant role for the observed cross-sector heterogeneity in profitability between energy and non-energy sectors (Konczal & Lusiani 2022). It is when we move to the joint analysis of energy price shifts, demand shifts and supply bottlenecks that we introduce a three-sector model (energy, goods, and services) for which we provide a mix of analytical discussion and numerical simulation based on empirically calibrated parameters.

Before we proceed, a number of remarks about scope and limitations are in order. First, our model focuses on short- to medium-run dynamics of inflation and income distribution rather than long-run growth. The capital stock and productivity are thus assumed to be predetermined.⁷ Second, we assume that the real wage targets of workers are exogenous. This assumption departs from standard wage-Phillips curve frameworks, in which economic activity feeds into the inflationary process via the bargaining power of workers. By contrast, in the three-sector version of our model, economic activity affects firms' price setting. This allows us to focus on the determination of mark-ups, which has been a key object of interest in the recent inflation debate, and to model an exogenous reduction in wage targets, e.g. by forcing workers to accept a lower real wage, as has been suggested in policy debates. That said, our model still exhibits the standard inflation-output trade-off, however without the existence of an exogenous NAIRU.

⁵Profit margins are equivalent to real unit profits which take the three-way distribution of sector revenues between capital, labour and intermediary inputs into account. They provide an absolute measure of capital remuneration in a given sector in the same way in which real (unit) wages provide a measure of labour's remuneration. For this reason profit margins are our preferred measure of sectoral profitability (see section 2.1.3)

⁶Diverging profit margins in response to cost shocks (such as energy price shocks) can arise for various reasons, including differences in market concentration or the shape of demand functions faced by different sectors (Bräuning et al. 2022). Recent survey data provided by Dogra et al. (2023) support the importance of demand conditions in driving the price-setting behaviour of firms, in line with our assumption of a capacity utilisation-elastic mark-up.

⁷For this reason, we also abstract from other, medium- to long-run drivers of rising markups or changes in income distribution, such as technology-induced network effects, rising market concentration or changes in bargaining institutions (Autor et al. 2020, Barkai 2020, De Loecker et al. 2020, Guschanski & Onaran 2022).

Third, for simplicity, households do not consume energy directly. Energy is typically provided to households through intermediaries such as gas stations or utilities, which themselves purchase energy from energy providers. Fourth, we model the persistent increase in global energy prices as a permanent shift rather than a one period shock. We see this as a reasonable simplification of the observed persistence of increased global energy prices. In the rest of the paper we refer to shifts in a variable or parameter as a permanent change. Fifth, our model does not include rational expectations of firms or workers. Instead we assume that workers pursue a standard of living which is given in the short run while firms pursue target profit margins. Both actors manage to change nominal wages and prices only with a delay.⁸ Finally, the model does not include fixed overhead costs of production that could mechanically lead to increasing profit shares in economic expansions (Lavoie 2023). We abstract from this mechanism because of the recent evidence of rising mark-ups which requires an explanation beyond fixed overhead costs.

2.1 A two-sector benchmark model

The two-sector benchmark model contains two sectors indexed as i = 1, 2, a domestic energy sector (sector 1) and a final output sector (sector 2).⁹ Energy is used as an intermediate input in sector 2. We begin by characterising output determination in the goods market, which clears instantaneously through quantity adjustment, before specifying the dynamics of workers' wagesetting and firms' price-setting behaviour, which pins down income distribution. The latter feeds back into economic activity via its impact on consumption demand. In the two-sector model there is no feedback yet from the goods market to wage and price setting, which will only be introduced in the three-sector extension below.

2.1.1 The goods market

Sector 1 produces the intermediate input (Y_1) , taken as energy.¹⁰ The production technology is such that each unit of sector 2 output (Y_2) requires δ units of intermediate input. Thus δ is the energy intensity of output in sector 2. The energy sector sets an exogenous real price for energy (ϵ) , which we discuss in detail in section 2.1.2.

$$Y_1 = \delta Y_2 \tag{1}$$

$$=\frac{p_1}{p_2}.$$
 (2)

Nominal value added (VA_i) is defined as revenues minus intermediate inputs, where p_i gives the price of the i^{th} sector's output:

 ϵ

$$VA_1 = p_1 Y_1 \tag{3}$$

$$VA_2 = p_2 Y_2 - p_1 Y_1. (4)$$

⁸Hein & Stockhammer (2010) provide a conflict inflation model with adaptive expectations.

⁹A full list of variable definitions and equilibrium solutions can be found in Appendix A.

 $^{^{10}}$ We drop all time subscripts in section 2.1.1 in order to avoid clutter. The dynamics of the model depend entirely on the interaction between wages and prices which are discussed in section 2.1.2, which is where we introduce time subscripts. This means for simplicity some variables have no time subscripts in section 2.1.1 but will have time subscripts in later sections.

Both sectors exhibit constant labour productivity (v_i) measured as physical units of output (Y_i) per worker employed (L_i) :

$$v_i = \frac{Y_i}{L_i}.$$
(5)

Nominal gross profits $(p_2\Pi)$ are defined as the difference between nominal value added and nominal wages paid. The aggregate wage sum (W) is the product of the sector's nominal wage (w_i) and the sector's employment (L_i) :

$$p_2\Pi = VA_1 - w_1L_1 + VA_2 - w_2L_2 \tag{6}$$

$$p_2 W = w_1 L_1 + w_2 L_2. (7)$$

Real final output (Y_2) is determined by consumption (C) and autonomous demand (X). The latter represents demand components other than consumption which are independent of changes in income such as discretionary government spending and autonomous investment.¹¹ Aggregate real income consists of the real wage bill (W) and aggregate profits (Π) :

$$Y_2 = C + X = W + \Pi.$$
 (8)

Agents are either workers or capitalists. The former receive wage income which is fully consumed and the latter receive profit income out of which a proportion (s) is saved. This two-class framework introduces a functional distribution of income and is a stylized representation of the well-documented higher propensity to consume out of wage compared to profit income (Onaran & Galanis 2014). The resulting aggregate consumption function is of the following form:

$$C = W + (1 - s)\Pi.$$
 (9)

Combining equations (6), (3), (4) and (5), (8) and (9) yields the following goods market equilibria:

$$Y_2^* = \frac{X}{sh} \tag{10}$$

$$Y_1^* = \frac{\delta X}{sh},\tag{11}$$

The aggregate profit share $h = \frac{\Pi}{Y_2}$ is given by:

$$h = 1 - \delta \frac{\omega_1}{v_1} - \frac{\omega_2}{v_2}.$$
 (12)

where ω_i are the sector real consumption wages ($\omega_i = \frac{w_i}{p_2}$). The sector profit shares are defined as nominal sector profits over nominal sector value added, $h_i = \frac{p_2 \Pi_i}{VA_i} = \frac{VA_i - w_i L_i}{VA_i}$, which yields

$$h_1 = 1 - \frac{\omega_1}{v_1 \epsilon} \tag{13}$$

$$h_2 = 1 - \frac{\omega_2}{v_2 \left(1 - \delta\epsilon\right)}.\tag{14}$$

¹¹There is a long tradition of modelling investment as endogenous to profit margins or capacity utilisation in post-Keynesian models (e.g. Hein (2014, chap. 5)). We adopt this simplifying assumption because our focus is on how energy price shifts affect distribution through price and wage-setting behaviour.

From equations (10) and (11) it can readily be seen that the profit share and output are inversely related, which reflects the higher marginal propensity to consume out of wage relative to profit income. Next we model how nominal wages and prices evolve as the result of wage bargaining and price setting by firms. This allows us to analyse how an exogenous increase in energy prices affects income distribution and output.

2.1.2 Wage and price setting

The energy sector sets an exogenous real price for energy (ϵ) :

$$\epsilon = \frac{p_{1,t}}{p_{2,t}}.$$
(15)

Modelling the price of energy as exogenous is motivated by the fact that a key component of energy prices is determined in global markets. Following the theoretical literature on commodity price shocks (Medina & Soto 2005, Sánchez 2008), we take the *real* energy price to be exogenous.¹² We further assume a constant real wage in the energy sector:¹³

$$\omega_1 = \frac{w_{1,t}}{p_{2,t}}.\tag{16}$$

In the non-energy sector, workers and firms try to influence real wages by setting nominal wages and prices, respectively. Our specification for conflict inflation builds on Blecker & Setterfield (2019, chap. 5)'s one-sector model, but adds a domestic intermediate input sector. Workers in the final output sector (sector 2) adjust nominal wages (w_2) with the aim of achieving a real wage target (ω_2^W) . We assume that the relevant aspiration of workers are real consumption wages (i.e., in prices of the final good $p_{2,t}$) rather than real product wages (i.e. in the price of the corresponding sector), because real consumption wages represent workers' purchasing power (see Aboobaker 2022). The bigger the gap between workers' aspirations (ω_2^W) and the actual real wage $(\omega_{2,t})$, the higher the increase in nominal wages.

$$w_{2,t+1} = w_{2,t} + w_{2,t}\phi_2\left(\omega_2^W - \omega_{2,t}\right)$$
(17)

Workers will only be able to partially realise their real wage target. Their adjustment speed depends on bargaining institutions, represented by parameter ϕ_2 in equation (17). Institutional factors influencing ϕ_2 include protective trade union legislation, bargaining coordination, union density, and collective bargaining coverage (Guschanski & Onaran 2022, Bhuller et al. 2022).

 $^{^{12}}$ If instead the *nominal* energy price would grow at an exogenous rate, relative prices would either go to infinity or to zero. Kilian & Zhou (2022b) suggest that the US consumer price index and the nominal gasoline price may be cointegrated, yielding a stationary real price. Our specification could be viewed as a simplified version of that assumption.

¹³Holding real consumption wages in the energy sector fixed is motivated by the fact that energy sector workers are uniquely positioned to defend their real wages when inflation rises due to energy price hikes. For example, the oil major Shell paid an 8% bonus to all staff in 2022.

Sector 2 firms set prices in pursuit of a target profit margin (r_2^T) . How quickly and how aggressively firms seek to close the gap between their target and the realised profit margin $(r_{2,t})$ is captured by parameter ψ_2 and depends on intra-sector competition as well as market regulation:

$$p_{2,t+1} = p_{2,t} + p_{2,t}v_2\psi_2\left(r_2^T - r_{2,t}\right).$$
(18)

Realised profit margins are defined as sector profits relative to sector sales $(\frac{\Pi_{i,t}}{Y_{i,t}})$, rather than relative to sector value added in the case of the profit share $(\frac{p_{i,t}\Pi_{i,t}}{VA_{i,t}})$. Another useful way of defining the profit margin of a sector is as the share of unit profits in unit sales. Unit profits are defined as the difference between price $(p_{i,t})$ and unit costs $(c_{i,t})$. The latter consist of unit labour costs $(w_{i,t}/v_i)$ and unit energy costs $(\delta p_{1,t})$, except for sector 1 which does not use energy as an input itself):

$$r_{2,t} = \frac{p_i \Pi_i}{p_i Y_i} = \frac{p_{2,t} - c_{2,t}}{p_{2,t}} = \frac{p_{2,t} - \frac{w_{2,t}}{v_2} - \delta p_{1,t}}{p_{2,t}} = 1 - \frac{\omega_{2,t}}{v_2} - \delta\epsilon$$
(19)

$$r_1 = 1 - \frac{\omega_1}{\epsilon v_1}.\tag{20}$$

In contrast to the sector profit share, which measures the distribution of value added between firms and workers within a sector and which will be discussed below, the profit margin is a measure of the profitability across sectors.

Equation (21) defines firms' target price as a mark-up over unit costs.¹⁴ The target profit margin (r_2^T) is then solely determined by the target mark-up:

$$p_{2,t}^T = (1+\theta^T)c_{2,t}$$
(21)

$$r_2^T = \frac{p_{2,t}^T - c_{2,t}}{p_{2,t}^T} = \frac{\theta^T}{1 + \theta^T}.$$
(22)

Together, equations (17) and (18) constitute a formal representation of conflict inflation. For example, an increase in prices due to an increase in firms' target mark-up will lead to a response by workers in the form of higher nominal wages, which in turn feeds into higher prices. As will be shown in the next section, this process will settle around a stable real wage and inflation rate.

2.1.3 Income distribution and inflation in equilibrium

The dynamic equations (17) and (18) admit an equilibrium that is characterised by a balanced nominal growth path in wages and prices, resulting in a stable real wage.¹⁵

¹⁴This mark-up pricing approach to internal planning is well established not only in the post-Keynesian literature (e.g. Hein 2014, chap. 5; Lavoie 2014, chap. 3) but also in Management Accounting (Drury 2017, Burns et al. 2013) and in practice (Drury & Tayles 2006).

¹⁵Since we assume constant labour productivity the real wage must be constant for a balanced growth path. Including growing labour productivity into the model would not alter the results substantially. See e.g. Hein (2014, chap. 8) and Blecker & Setterfield (2019, chap. 5.2) for one-sector models with labour productivity growth.

Proposition 1 With prices and nominal wages growing at the same rate $\hat{p}_2 = \hat{w}_2$, the equilibrium real wage and inflation rate are given by:

$$\omega_2^* = \frac{\phi_2 \omega_2^W + \psi_2 v_2 \left(1 - r_2^T - \delta\epsilon\right)}{\psi_2 + \phi_2} \tag{23}$$

$$\hat{p}_{2}^{*} = \frac{\psi_{2}\phi_{2}}{\psi_{2} + \phi_{2}} \left[\omega_{2}^{W} - v_{2} \left(1 - r_{2}^{T} - \delta \epsilon \right) \right].$$
(24)

Using sector 2 firms' internal markup pricing approach (21) we can define the expression $v_2\left(1-r_2^T-\epsilon\delta\right)$ as firms' implicit desired real wage ω_2^F . ¹⁶ The equilibrium real wage is thus a weighted average of workers' actual and firms' implicit real wage targets (ω_2^W and ω_2^F), with weights given by the nominal wage and price adjustment speeds (ϕ_2 and ψ_2). A strictly positive equilibrium real wage requires the terms r_2^T and $\epsilon\delta$ to be sufficiently small. Economically this means the real energy price, the energy intensity of production, and the target mark-up cannot become too high.

The larger the gap between workers' and firms' real wage targets, the higher the inflation rate. Bargaining power also affects inflation through the two adjustment speeds, if firms (or workers) manage to adjust prices (or nominal wages) faster, the result is not only higher inflation but also a lower (higher) real wage.

We can use these equilibria to distinguish between wage-price and price-wage spirals. While both lead to higher inflation, their distributional effects are different. A wage-price spiral is triggered by an increase in ω_2^W or ϕ_2 and lead to an increase in the equilibrium real wage ω_2^* . In contrast a price-wage spiral is triggered by an increase in r_2^T , ψ_2 or ϵ and leads to a lower real wage in addition to a higher equilibrium inflation rate.

Proposition 1 requires prices and wages to grow at the same rate. Proposition 2 derives the conditions under which this is likely to occur (see Appendix C for a discussion). A positive equilibrium inflation rate requires that workers' real wage target exceeds firms' implicit real wage target ($\omega_2^W > \omega_2^F$) and that the energy price or firms' target profit margin does not become too large relative to workers' real wage target.

Proposition 2 If workers' real wage target exceeds firms' implicit real wage target $(\omega_2^W > \omega_2^F)$ and $\phi_2 \omega_2^W + \psi_2 \omega_2^F > 0$, then the system given by (17)–(18) is likely to exhibit exponential growth with equal growth rates for $w_{2,t}$ and $p_{2,t}$ and positive equilibrium solutions for the real wage and inflation rate.

Proof. See Appendix C.

Consider an exogenous increase in the real energy price (ϵ). This will increase the gap between target and realised profit margins in sector 2 and therefore will negatively affect equilibrium real

¹⁶It is easy to show that $\omega_{2,t} - \omega_2^F = v_2(r_2^T - r_{2,t})$. Thus firms' price setting equation could also be written in terms of ω_2^F .

wages:

$$\frac{\partial \omega_2^*}{\partial \epsilon} = -\frac{v_2 \psi_2 \delta}{\psi_2 + \phi_2} < 0.$$
⁽²⁵⁾

Given that energy sector workers fully defend their real wage by equation (16), whereas real wages in the final output sector fall, an energy price shift thus leads to changes in the between-sector wage distribution.

Furthermore, by equation (24), the shift in energy prices will increase the equilibrium inflation rate. The size of this effect depends again on the combined bargaining power as well as on the energy intensity of the final output sector (δ) and labour productivity (v_2):

$$\frac{\partial \hat{p}_2^*}{\partial \epsilon} = \frac{\psi_2 \phi_2 v_2 \delta}{\psi_2 + \phi_2} > 0.$$
(26)

Next, we examine the distributional impact of energy prices on the aggregate and sector profit shares as well as profit margins, whose equilibrium expressions are given by:

$$h^* = 1 - \frac{\omega_2^*}{v_2} - \delta \frac{\omega_1}{v_1} \tag{12}$$

$$r_1 = 1 - \frac{\omega_1}{\epsilon v_1} = h_1 \tag{13}$$

$$r_2^* = 1 - \frac{\omega_2^*}{v_2} - \delta\epsilon \neq h_2^*$$
(27)

$$h_2^* = 1 - \frac{\omega_2^*}{v_2} \frac{1}{1 - \delta\epsilon}.$$
 (14)

For sector 1 the realised profit margin is equal to the profit share. This follows directly from the fact that no intermediate inputs are used in sector 1. In contrast in sector 2 unit sales are divided between wages, profits and intermediate inputs. As a result the distribution of income (value added) within the sector as measured by the profit share (h_2^*) does not coincide with the profit margin (r_2^*) . Thus, sector 2 profit margins can in principle react differently to an energy price shock than profit shares. It is therefore important to separately analyse income distribution within sectors (measured by the profit shares) and profitability between sectors (measured by the profit margin).

Both the aggregate and sector 1 profit share unambiguously increases in response to a positive shift in the price of energy (see equations (28) and (29)). By contrast, in sector 2 the increase in energy prices leads to a reduction in profit margins (see equation 30) which is greater for higher energy intensity (δ) and higher worker's bargaining power ($\phi_2/(\phi_2 + \psi_2)$). Thus changes in the real price of energy increase heterogeneity in profitability between sectors.

$$\frac{\partial h^*}{\partial \epsilon} = \frac{\delta \psi_2}{\phi_2 + \psi_2} > 0 \tag{28}$$

$$\frac{\partial h_1}{\partial \epsilon} = \frac{\partial r_1}{\partial \epsilon} = \frac{\omega_1}{v_1 \epsilon^2} > 0 \tag{29}$$

$$\frac{\partial r_2^*}{\partial \epsilon} = -\frac{\delta \phi_2}{\psi_2 + \phi_2} < 0 \tag{30}$$

A key result is the effect of the energy price shift on the profit share in sector 2:

Proposition 3 The response of the profit share in sector 2 (h_2^*) to a shift in real energy prices is ambiguous. It depends on the relative strength of the bargaining relationship, specifically on the relative size of ϕ_2 and ψ_2 , and the gap between firms' target profit margin (r_2^T) and workers real wage target (ω_2^W) .

Proof.

$$\frac{\partial h_2^*}{\partial \epsilon} = \delta \frac{\psi_2 r_2^T - \phi_2 \frac{\omega_2^W}{v_2}}{\left(1 - \epsilon \delta\right)^2 \left(\psi_2 + \phi_2\right)} \leq 0.$$
(31)

Thus, it is not a priori clear which class within sector 2 bears how much of the adjustment burden in response to an increase in energy prices. The outcome depends on bargaining power: if the price-setting power of firms is sufficiently large, profit shares of domestic non-energy sectors may *rise* in response to an energy cost shock. This notable result is closely linked to the recent debate on rising profits in response to adverse economic shocks (Konczal & Lusiani 2022, Weber & Wasner 2023, Unite 2022). It shows that with conflict inflation, the redistribution of income from workers to firms due to a pure energy price shocks may not be limited to energy firms but can occur more widely across different sectors of the economy. By contrast, a pure energy price shock cannot explain an increase in profit *margins* in non-energy sectors.

The decline in real wages due to an increase in the real energy price increases the aggregate profit share, which depresses consumption demand and thus output:

$$\frac{\partial Y_2^*}{\partial \epsilon} = -\left(\frac{X}{sh^2}\right) \left(\frac{\psi_2 \delta}{\phi_2 + \psi_2}\right) < 0 \tag{32}$$

$$\frac{\partial Y_1^*}{\partial \epsilon} = -\left(\frac{X}{sh^2}\right) \left(\frac{\psi_2 \delta^2}{\phi_2 + \psi_2}\right) < 0.$$
(33)

With constant labour productivity (and a fixed labour force in the short run), the fall in output will be accompanied by an increase in unemployment.

The causal structure of the model is summarised in the directed graph in Figure 2.¹⁷ An increase in the real price of energy (ϵ) leads to a rise in final output prices (p_2), a fall in the sector 2 real wage, and thus a change of the wage distribution. Workers in the energy sector retain their living standard but obtain a smaller share of the sector's output (the profit share increases). Profit margins increase in sector 1 and drop in sector 2, however the effect on the distribution of income between labour and capital in sector 2 is ambiguous and depends on the relative strength of workers and firms in the bargaining process. However, the aggregate profit share rises, which depresses consumption, output, and employment.

¹⁷Exogenous factors are represented by nodes with only outward-pointing arrows. Other exogenous parameters (e.g. labour productivity, energy intensity, propensity to save out of profits) are omitted for simplicity.





2.2 A three sector extension

While the two-sector model captures the redistribution of income from workers to firms, it fails to fully account for the divergence in profit margins across non-energy firms which occurred since the pandemic (Hayes & Jung 2022, Konczal & Lusiani 2022, Weber & Wasner 2023). We account for this by disaggregating final output into goods and services, yielding a three-sector model. This allows us to capture supply bottlenecks across various industries as well as the significant shift in demand away from services towards goods that occurred during the pandemic due to lockdowns and physical distancing. The combination of a capacity utilisation-elastic mark-up in the goods sector and sector-specific demand and supply shocks allows us to replicate the empirically observed divergence in profit margins across non-energy sectors (Hayes & Jung 2022). The utilisation-sensitive mark-up represents the idea that firms tend to increase prices when they struggle to meet demand (Flaschel & Skott 2006, Weber & Wasner 2023).¹⁸

¹⁸Weber & Wasner (2023, p. 10) argue that 'where a sector furthermore experiences a supply-side bottleneck or a demand shock – granting firms within the sector temporary augmented monopoly power – profit margins may even be enhanced, thereby not only propagating but also amplifying the initial cost shocks'. Our model can

The endogenous mark-up further introduces a feedback effect from the goods market into the inflationary process that gives rise to a relationship that is similar to the traditional Phillips curve, but runs via price- rather than wage-setting.

2.2.1 The goods market

The three sector model consists of two final outputs, services (sector 2) and goods (sector 3), which are used for consumption (C_n) and autonomous components of demand (X_n) such as government spending (the subscript *n* refers to nominal quantities). Final output (Y_n) is equal to total income which consists of the wage bill (W_n) and aggregate profits (Π_n) . Equality between aggregate income and expenditures as well as the aggregation of final output is defined in nominal terms:¹⁹

$$Y_n = p_2 Y_2 + p_3 Y_3 = C_n + X_n = \Pi_n + W_n \tag{34}$$

Both final output sectors use inputs from sector 1 (energy) in a fixed but sector-specific proportion (δ_i) representing the energy intensity of sector *i* which is defined based on real quantities:

$$Y_1 = \delta_2 Y_2 + \delta_3 Y_3.$$
 (35)

The definitions of aggregate profits, wages, and value added are equivalent to those introduced in section 2.1.1 and are therefore omitted for brevity. We use the same consumption function as before (equation (9)). Households spend a fixed proportion of their nominal income on goods and services, respectively, yielding a fixed ratio (γ) of nominal sector 2 output to total final output:

$$\gamma = \frac{p_2 Y_2}{Y_n}.\tag{36}$$

The aggregate price level is the weighted geometric mean of final output prices, using the fixed proportions of final output as weights:

$$p = p_2^{\gamma} p_3^{1-\gamma}.$$
 (37)

We use the aggregate price index to define real quantities such as real autonomous demand $X = \frac{X_n}{p}$ and real aggregate output $Y = \frac{Y_n}{p}$. Relative prices are define as $\mu = \frac{p_2}{p_3}$. The fixed nominal proportion γ in combination with fixed sector 1 input coefficients yield the real output proportions $Y_2 = \frac{\gamma}{(1-\gamma)\mu}Y_3$ and $Y_1 = \left(\delta_2 \frac{\gamma}{(1-\gamma)\mu} + \delta_3\right)Y_3$. By the same procedure as in section 2.1.1, we obtain the following goods market equilibrium expressions for nominal and real output:

$$Y_n^* = \frac{Xp}{sh} \tag{38}$$

be seen as a formalisation of this argument.

¹⁹As in the two sector model, we drop all time subscripts in section 2.2.1 in order to avoid clutter. The dynamics of the model depend entirely on the interaction between wages and prices which are discussed in section 2.2.2, which is where we introduce time subscripts. This means some variables have no time subscripts in section 2.2.1 but will have time subscripts in later sections.

$$Y^* = \frac{X}{sh} \tag{39}$$

The aggregate profit share is given by real unit labour costs in all three sectors (ω_i/v_i) and the relative size of the sectors which in turn depend on the economy's energy intensity (δ_i) , the composition of final demand (γ) and relative final output prices (μ) :

$$h = 1 - (1 - \gamma)\mu^{\gamma} \left(\frac{\omega_3}{v_3}\right) - \gamma \mu^{\gamma - 1} \left(\frac{\omega_2}{v_2}\right) - \left[\gamma \delta_2 \mu^{\gamma - 1} + (1 - \gamma)\delta_3 \mu^{\gamma}\right] \left(\frac{\omega_1}{v_1}\right).$$
(40)

Sector profit shares are defined as nominal sector profits relative to sector value added $h_i = \frac{p_i \Pi_i}{V A_i}$. They are directly related to real wages $\omega_i = w_i/p$, the real price of energy, $\epsilon = \frac{p_1}{p}$, and the relative price of the two final outputs, μ . As in the two sector model the real wage in the energy sector (ω_1) is assumed to be exogenous.

$$h_1 = 1 - \frac{w_1}{p_1} \frac{1}{v_1} = 1 - \frac{\omega_1}{\epsilon v_1} \tag{41}$$

$$h_2 = 1 - \frac{w_2}{v_2 \left(p_2 - p_1 \delta_2\right)} = 1 - \frac{\omega_2}{v_2 \left[(\mu)^{1 - \gamma} - \epsilon \delta_2\right]}$$
(42)

$$h_3 = 1 - \frac{w_3}{v_3 \left(p_3 - p_1 \delta_3\right)} = 1 - \frac{\omega_3}{v_3 \left[\mu^{-\gamma} - \epsilon \delta_3\right]} \tag{43}$$

2.2.2 Wage and price setting

This section introduces wage bargaining and price setting. Following the two-sector benchmark model, we keep the real energy price and the real wage in the energy sector fixed, using the price index as the deflator:

$$\epsilon = \frac{p_{1,t}}{p_t} \tag{44}$$

$$\omega_1 = \frac{w_{1,t}}{p_t}.\tag{45}$$

Workers set nominal wages depending on how far actual real wages $(\omega_{i,t})$ are from workers' target real wage (ω_i^W) :

$$w_{2,t+1} = w_{2,t} + w_{2,t}\phi_2\left(\omega_2^W - \omega_{2,t}\right)$$
(46)

$$w_{3,t+1} = w_{3,t} + w_{3,t}\phi_3\left(\omega_3^W - \omega_{3,t}\right) \tag{47}$$

Firms in both sectors set prices depending on the gap between the realised $(r_{i,t})$ and target profit margin $(r_2^T \text{ and } r_{3,t}^T)^{20}$

$$p_{2,t+1} = p_{2,t} + p_{2,t}\psi_2 v_2 \left(r_2^T - r_{2,t}\right)$$
(48)

$$p_{3,t+1} = p_{3,t} + p_{3,t}\psi_3 v_3 \left(r_{3,t}^T - r_{3,t}\right)$$
(49)

Realised profit margins $r_{i,t} = \frac{p_{i,t}-c_{i,t}}{p_{i,t}} = 1 - \frac{w_{i,t}}{p_{i,t}v_i} - \frac{\delta_i p_{1,t}}{p_{i,t}}$ depend on prices relative to real unit costs $(c_{i,t}/p_{i,t})$. The latter can be broken down into real unit labour costs $(\frac{w_{i,t}}{p_{i,t}v_i})$ and real energy costs $(\frac{\delta_i p_{1,t}}{p_{i,t}})$.

²⁰In line with Aboobaker (2022) this implies that workers care about the real consumption wage $\omega_{i,t} = \frac{w_{i,t}}{p_t}$, whereas for firms the real product wage $\frac{w_{i,t}}{p_{i,t}}$ is key.

The key difference between sector 2 (services) and sector 3 (goods) lies in the determination of the target mark-up, which is exogenous in sector 2 but endogenous in sector 3. This simplified assumption captures established evidence that goods (and especially durables) change prices more frequently in line with the business cycle than services (Klenow & Malin 2010). Following Flaschel & Skott (2006), we assume that the target mark-up in sector 3 reacts to the rate of capacity utilisation given by $\frac{Y_{3,t}}{Y_3^P}$, where Y_3^P is potential output (taken to be exogenous in the short run):

$$\theta_{3,t}^T = \beta_0 + \beta_1 \left(\frac{Y_{3,t}}{Y_3^P}\right) \tag{50}$$

In equilibrium, $Y_3 = (1 - \gamma) \mu^{\gamma} Y$; thus mark-ups in sector 3 can change both due to aggregate (Y) and sector demand shocks (γ) , as well as supply shocks, e.g. bottlenecks, represented by temporary exogenous changes to potential output (Y_3^P) .

Target profit margins are given by the target mark-ups as before: $r_i^T = \frac{\theta_i^T}{1+\theta_i^T}$. The realised profit margins become:

$$r_1 = h_1 \tag{51}$$

$$r_2 = 1 - \mu^{\gamma - 1} \frac{\omega_2}{v_2} - \mu^{\gamma - 1} \delta_2 \epsilon \neq h_2$$
(52)

$$r_3 = 1 - \mu^{\gamma} \frac{\omega_3}{v_3} - \mu^{\gamma} \delta_3 \epsilon \neq h_3 \tag{53}$$

Profit margins thus do not coincide with profit shares for sectors which use intermediary inputs.

Finally, it's worth discussing how the distinction between the goods and services sector affects our analysis. By adding a domestic energy sector to the model, the conflict over the distribution of income expanded from one dimension (firms vs workers) to two dimensions (energy firms vs final goods firms). With two final output sectors, the firm vs firm dimension now also includes a distributional conflict between the two final output producers. This latter point stems from the fact that both input costs, nominal wages w_i and the nominal energy price p_1 , react to changes in the aggregate price level (p_t) . Thus, if sector *i* manages to increase prices at a faster pace than the aggregate price level, its profit margin increases permanently even if it achieves above-average price growth only for a limited period. Overall, the three sector model provides a rich framework for analysing cost shocks and anti-inflationary policies and their implications for distributional as well as macroeconomic outcomes. Crucially it allows us to study not only the conflict between firms and workers within a sector but also between firms across sectors.

2.2.3 Income distribution and inflation in equilibrium

The system given by (46)–(50) is likely to exhibit balanced growth in nominal wages and prices over a wide range of parameters (see Appendix D). While prices and nominal wages are unstable (i.e. grow over time), the system converges to stable equilibrium real wages that are below workers' targets (see Appendix E). The causal structure of the model is summarised in Figure $3.^{21}$ The crucial determinants of inflation and distribution on which we focus in this model are the

²¹Some exogenous factors such as energy intensity and labour productivity are omitted for simplicity.

exogenous real energy price (ϵ) , the share of goods in total output $(1-\gamma)$, and potential output in the goods sector (Y_3^P) . These factors determine sectoral price setting via their impact on costs and the target mark-up in sector 3. Sectoral prices then determine the aggregate price level and real wages, which feed back into the inflationary process via wage-setting. Together these determine the functional as well as inter sector distribution of income. The former determines equilibrium output, which feeds back into the bargaining process.





Analogous to the two sector model we can distinguish wage-price and price-wage spirals. Both spirals lead to higher equilibrium inflation but with distinct distributional outcomes. A wage-price spiral in sector i is triggered by an increase in ω_i^W or ϕ_i and leads to an increase in the equilibrium real wage ω_i^* . In contrast, a price-wage spiral is triggered by an increase in r_i^T , ψ_i or ϵ and leads to a lower real wage.

3 Empirical calibration

To analyse the effects of policy interventions, we complement the analytical discussion with numerical simulations. We calibrate 16 of the model's parameters with the Method of Simulated Moments (MSM) using sector data for the US. The MSM calibrates parameters so as to minimise the distance between simulated and empirical moments (Franke & Westerhoff 2012, Franke 2022, Reissl 2020). In this way, we aim to obtain more realistic simulation results compared to an arbitrary parameterisation. However, a number of limitations needs to be flagged. First, the model is relatively small-scale and will thus omit certain macroeconomic processes that impact the empirical series. Second, our model does not allow for any parameter instabilities in the data-generating process, e.g. due to changes in institutions or technology. Correspondingly, we do not aim to maximise the model's empirical fit, nor do we consider its parameters to be strictly causally identified. Instead, the purpose of the calibration is to incorporate empirical information into an analysis that is essentially theoretical. Thus, any counterfactual policy analysis should be considered a theoretical exercise to gauge possible responses rather than a quantitatively precise prediction.

For the calibration, we use annual time series at the two digit NAICS sector level for the US over the period 2000-2021.²² We construct tree sectors: a domestic energy sector (S1), a services sector (S2), and a goods sector (S3) (see Appendix F for details on the data).

We construct data for (i) sector real outputs,²³ (ii) the inflation rates of sector price indices, (iii) the sector profit shares, and (iv) an energy price index that is deflated by the price indices of the goods and service sectors. From these series, we obtain 18 empirical moments 24

The stochastic components used to obtain the simulated counterparts to these empirical moments are twofold. First, we directly feed the empirical real energy price index into the model. Second, we add an autocorrelated aggregate demand shock u_t to equation (34), which follows the process $u_t = \rho u_{t-1} + e_t$, with $e_t \sim N(0, \sigma)$. We then run 30 Monte Carlo simulations of the stochastic version of the model over a time horizon H = 16T (with T = 22), discard H/2 initial periods, and then compute the simulated moments based on the mean values across Monte Carlo runs.

Having obtained q = 18 empirical moments and their simulated counterparts, we calibrate p = 16 parameters, which are listed in Table 1. Let the $(p \times 1)$ vector of parameters to be calibrated

²²Data availability constraints restrict the latest data point to 2021 and preclude the use of quarterly data.

²³The real output series were detrended using the regression filter proposed in Hamilton (2018).

 $^{^{24}}$ The contemporaneous cross-correlations between the real energy price index and (i) the sector price indices (moments 1 - 3), (ii) the profit shares in S2 and S3 (moments 4 - 5), and (iii) real final output (sum of real output in S2 and S3) (moment 6). Furthermore, we use the cross-correlation of the inflation rate in S1 and (i) the profit share in S1 (moment 7) and (ii) the final output inflation rate (growth rate of price index of S2 and S3) (moment 8). To capture persistence in the data, we use the first-order autocorrelation coefficients of the aggregate inflation rate (moment 9), of the profit shares in S2 and S3 (moments 10 - 11), and of real final output (moment 12). Finally, we use the mean of the final output inflation rate (moment 13) as well as the standard deviations of (i) the inflation rates in S2 and S3 (moments 14 - 15), (ii) the profit shares in S2 and S3 (moments 16 - 17), and (iii) real final output (moment 18).

be denoted as Θ , the vector of empirical moments m^{emp} , and the vector of simulated moments by m^{sim} . The MSM then consist of minimising the following quadratic loss function:

$$L(\Theta) = (m^{sim}(\Theta) - m^{emp})'W(m^{sim}(\Theta) - m^{emp}),$$
(54)

where W is a $(q \times q)$ weighting matrix. Following Franke & Westerhoff (2012), we use the inverse of the bootstrapped variance-covariance matrix of m^{emp} to construct W. The loss function is then minimised using the parameter vector Θ as the choice variable:

$$\hat{\Theta} = \underset{\Theta}{\arg\min[L(\Theta)]}.$$
(55)

To find a parameter vector that minimises the loss function, we follow Reissl (2020) and use Latin hypercube sampling by drawing S samples from a predetermined parameter space to construct a $(S \times p)$ parameter grid. We set S = 10000, run Monte Carlo simulations for each parameter combination, compute the simulated moments, and calculate the corresponding value of the loss function.²⁵ This procedure is repeated several times to narrow down the parameter range. We then choose the parameterisation that yields the smallest value of the loss function (see Table 1).

\mathbf{Symbol}	Definition	Pre-determined range	Calibrated value
θ_2	Mark-up, S2	(0.4, 1)	0.73
β_0	Autonomous component of mark-up, S3	(0.1, 0.4)	0.19
β_1	Sensitivity of mark-up to rate of capacity utilisation, S3	(0.1, 0.9)	0.31
ϕ_2	Adjustment speed of nominal wage, S2	(0.5, 1.5)	0.71
ϕ_3	Adjustment speed of nominal wage, S3	(0.5, 1.5)	0.55
ψ_2	Adjustment speed of price, S2	(1.8, 1.8)	1.59
ψ_3	Adjustment speed of price, S3	(1.8, 1.8)	1.76
ω_1	Real wage, S1	(0.1, 0.8)	0.44
ω_2^W	Real wage target of workers, S2	(0.25, 0.65)	0.25
ω_3^W	Real wage target of workers, S3	(0.25, 0.65)	0.57
8	Propensity to save	(0.2, 0.8)	0.64
Y_3^P	Potential output, S3	(1, 4)	2.19
ρ	AR(1) coefficient of demand shock	(0.1, 0.8)	0.39
σ	Standard deviation of demand shock	(0.1, 0.4)	0.19
δ_2	Energy coefficient, S2	(0.2, 0.8)	0.36
δ_3	Energy coefficient, S3	(0.2, 0.8)	0.51

Table 1: Empirically calibrated parameters

Notes: S1: sector 1 (energy), S2: sector 2 (services), S3: sector 3 (goods). The parameter γ was calibrated directly from the sector data to $\gamma = 0.55$. The labour coefficients v_i were normalised to unity. Autonomous demand was set to X = 3.

4 Simulation results

Using the empirically calibrated parameters, we analyse the model's dynamics under five different scenarios. The first scenario (Scenario 1) is a permanent increase in real energy prices. The

²⁵We discard parameterisations for which $\hat{p}^* < 0$ or $cor(\epsilon, h_2) < 0$ given that a positive inflation rate and a positive correlation between the energy price shock and profit shares in S2 are key empirical facts we want to capture.

second scenario (Scenario 2) adds to the increase in energy prices a shift of demand towards goods (sector 3) as well as a reduction in potential output (Y_3^P) to capture pandemic-induced disruptions. We consider this our baseline scenario. Subsequently, we analyse the effects of three policy interventions: first, a windfall tax on energy sector profits which is used to compensate workers in the other sectors. Such a tax has been introduced in the UK, Italy, Spain and Greece and debated in many other countries (e.g. Austria, Germany, France, and Belgium; see Enache 2022), but not been implemented in the US. Second, a policy of wage restraint in an attempt to reduce inflation, as called for by e.g. Domash & Summers (2022) (and e.g. Bailey (2022) and Hunt (2022)). We model this as an exogenous decrease in workers' real wage target (ω_2^W). Third, a reduction in aggregate demand, e.g. through contractionary fiscal or monetary policy, represented by a fall in autonomous demand X. We will compare these policies with respect to their effect on inflation, output (unemployment), and the distribution of income.

4.1 Scenario 1: Energy price shift

The first scenario consists of a permanent increase in the real price of energy (ϵ). The model's results are plotted in Figure 4. As expected a permanent increase in ϵ leads to an increase in the sectoral inflation rates (\hat{p}_i) (panel a). Panel (b) shows that aggregate prices start to rise before nominal wage inflation picks up, demonstrating that an energy price shock triggers a price-wage spiral. Real wages fall in sectors 2 and 3, but stay constant in the energy sector (panel c). The increased price of energy boosts the energy sector's profit margin as well as its profit share (panels d and e). In sector 2 and 3, profit margins fall while at the same time the profit shares increase, implying that worker's bear a bigger share of the increase in energy prices compared to firms. The aggregate profit share increases, indicating that over the whole economy firms' share of total income increases. Finally, output falls across all three sectors (panel f) due to the fall in consumption demand.²⁶ Overall, the new equilibrium is characterised by lower real wages and reduced profit margins outside the energy sector, and higher profit shares in most sectors and in the aggregate. Workers lose out not only in relative terms (profit shares) but also in absolute terms due to declining real wages (sectors 2 and 3) and through a fall in employment due to falling output (in all sectors).

4.2 Scenario 2 (baseline): Energy price shift combined with sectoral demand shift and supply disruption

Our baseline scenario involves (i) an increase in the real energy price (ϵ) , (ii) an increase in the share of goods in total output $(1 - \gamma)$, (iii) and a reduction in potential output of the goods sector (Y_3^P) . The results are displayed in Figure 5. Compared to the pure energy price shock scenario 1, we obtain a stronger increase in the sectoral inflation rates (panel a), with a much higher spike in inflation in the goods sector. Again, there is a price-wage spiral as prices start to increase before wage inflation picks up, with a higher peak inflation rate as well as a higher

²⁶Sector 3 output falls the most because of the temporary higher inflation compared to sector 2 (due to a higher adjustment speed ψ_3 in sector 3) which means the fixed proportion of nominal household income spent on sector 3 translates into lower demand for sector 3 in real terms.



Figure 4: Scenario 1: Energy price (ϵ) shift

Notes: Response to a permanent increase in real energy prices in period 10. Responses in panels c and f are normalized by their initial steady state values. Panels d and e show the change in profit margins and shares from the initial steady state.

equilibrium inflation rate compared to scenario 1 (panel b). Real wages in sectors 2 and 3 fall more compared to scenario 1 (panel c). The demand shift towards sector 3 in combination with supply bottlenecks leads to an increase in sector 3's target markup, which translates into higher profit margins (panel d). By contrast, sector 2's profit margins are squeezed. With respect to the distribution of income between labour and capital within sectors, the profit share increases in all sectors and in the aggregate (panel e). Real output again falls across sectors (panel f).

In sum, the combined shock scenario not only leads to an increase in aggregate inflation and a fall in aggregate output, it also comes with an increased variation of profit margins across sectors. Crucially this increase in heterogeneity is not only driven by energy sector profits but also by increased profitability in the goods sector. Thus the defining features of the recent macroe-conomic environment, high energy prices, pandemic-related supply bottlenecks and changes in consumer demand patterns, contribute to increasing differences in profitability across sectors. Sectors that are able to not just maintain but to increase their target mark-ups in response to shifts in demand or supply constraints manage to raise their profit margins in an overall adverse macroeconomic environment as highlighted in Weber & Wasner (2023). This increase in target markups further amplifies aggregate inflation. Between firms, the combined shocks produce winners (sector 1 and 3) and losers (sector 2) measured in terms of profit margins. The



Figure 5: Scenario 2 (baseline): Energy price (ϵ) , demand composition (γ) and supply bottleneck (Y_3^P) shifts

Notes: Responses to a joint permanent increase in real energy prices (ϵ) , the demand for goods $(1 - \gamma)$, and a reduction in potential output in sector 3 (Y_P^3) in period 10. Responses in panels c and f are normalized by their initial steady state values. Panels d and e show the change in profit margins and shares from the initial steady state.

increase in inflation redistributes income towards firms through falling real wages. As a result, workers lose out across all sectors. Sector 1 workers enjoy stable real wages but obtain a smaller share of valued added produced in their sector, while workers in the final output sectors lose out in absolute terms (falling real wages) as well as in relative terms (increasing profit shares). Workers in all sectors also suffer from lower employment.

4.3 Policy Scenario 1: Corporate income tax and income transfer

Consider first the introduction of a corporate income tax on the energy sector (Figure 6).²⁷ In keeping this scenario as simple as possible, we assume that a proportional tax on corporate profits in sector 1 is introduced and the revenues are immediately paid out as a lump sum to workers in sectors 2 and 3, keeping the government's budget balanced (see Appendix A for the details of the model extension). The corporate income tax-financed transfer to workers is effective in reducing inflation rates (panel a). The transfer increases workers' real disposable income, which is defined as wage plus transfer income (panel b). As a result, workers moderate their

²⁷Note that all policy scenarios start from a pre-inflationary shock scenario.

nominal wage demands and the economy settles at a lower inflation rate (panel c). The increase in worker's disposable income further leads to an output expansion via higher consumption demand (panel d), which translates into a rise in employment. Lower nominal wage growth leads to an increase in sector 2 firms' profit margin. Profit margins also increase in sector 3 due to the increased mark up (panel e). Finally, profit shares in sectors 2 and 3 increase slightly, while the aggregate (post-tax) profit share declines.





Notes: Response to the introduction of corporate income tax in the energy sector in period 10 (pre-inflation shock scenario). Responses in panels (b) and (d) are normalized by their initial steady state values. Panels (b) and (c) show workers' disposable income, defined as wages plus transfer receipts.

Overall, introducing a corporate income tax in the energy sector proves to be an effective tool for reducing inflation while also stabilising workers' disposable income and employment. The key mechanism through which this policy works is the transfer paid to households which helps soften the conflict over the distribution of income.

4.4 Policy Scenario 2: Wage restraint

The second scenario is a wage-restraint policy where workers in the sector which undergoes a fall in demand (sector 2), lower their real wage target (ω_2^W) , i.e. accept a fall in living standards (Figure 7). Convincing or forcing workers to halt nominal wage demands does lead to a fall in aggregate inflation (panel a) but also to a fall in output (panel b). Real wages decline for sector

2 workers but increase for sector 3 workers (panel c). The mechanism at play is that workers in sector 3 lower their nominal wage demand more slowly than the aggregate inflation falls (panel d). Wage growth below sector price growth allows both sectors to increase their profit margin (panel e), but only sector 2 manages to claim a larger share of sector value added (panel f), which however is large enough to also increase the aggregate profit share.



Figure 7: Policy Scenario 2: Sector 2 wage restraint

Notes: Responses to a permanent reduction in workers' real wage target (ω_2^W) in period 10. Responses in panels (b) and (c) are normalized by their initial steady state values. Panels (e) and (f) show the change in profit margins and shares from initial steady state values.

Overall, while effective in reducing inflation, a policy of wage restraint shifts the distribution of income towards firms and towards those workers who are able to maintain their real wage targets. Firms in those sectors where workers do engage in wage restraint (the services sector in our scenario), experience a boost in their profitability and a higher profit share. In the context of the current macroeconomic environment, a policy of sectoral wage restraint amounts to exacerbating inter-sectoral wage inequality as well as further depressing real wages of workers in that sector. Thus, confronting elevated inflation rates by wage restraint may curb inflation, but shifts the burden heavily on those workers who are not able to defend their real wages. It further boosts profit incomes which tend to be concentrated at the top of the income distribution.

4.5 Policy Scenario 3: Contractionary macroeconomic policy

The last policy scenario is a reduction in autonomous aggregate demand (X), e.g., via a cut to fiscal expenses or contractionary monetary policy (Figure 8). The reduction in aggregate demand leads to a fall in aggregate inflation as well as output and employment (panels a and b). The inflation rate falling faster than nominal wage growth, leads to an increase in real wages in sectors 2 and 3 (panels c and d). The fall in output leads to a reduction of the mark-up in sector 3 and as a result to lower profit margins (panel e). The aggregate profit share falls as well as in sector 3 (panel f).





Notes: Responses to a permanent reduction in autonomous demand (X) period 10. Responses in panels (b) and (c) are normalized by their initial steady state values. Panels (e) and (f) show the change in profit margins and shares from initial steady state values.

Thus, contracting aggregate demand leads to a reduction in inflation, but is associated with significant costs in the form of lower output and employment. Our analysis thus identifies major shortcomings in the form of adverse distributional side-effects of macroeconomic policies that aim to bring down inflation via wage-restraint or demand-contraction.

5 Conclusion

The paper proposes a three-sector model of conflict inflation to analyse the distributional effects of energy price shocks in economies with domestic energy production. We use this model to

study the recent experience of energy price increases combined with a pandemic-induced shift in consumer spending towards goods, and the occurrence of supply bottlenecks. Our main results, based on a calibration of the model to US data, are as follows. Firstly, energy price shocks allow energy firms to increase both their profit margins and profit shares, while in non-energy sectors only profit shares increase. Workers in the non-energy sectors get squeezed by falling real wages. Importantly, energy price shocks alone cannot explain increasing profit margins outside the energy sector in our framework. Second, to explain the empirically observed increased profit margins outside the energy sector (Bräuning et al. 2022, Hayes & Jung 2022, Konczal & Lusiani 2022, Weber & Wasner 2023), pandemic-induced demand shifts and supply bottlenecks in the goods sector need to be considered. In this context, our model is able to reproduce recent empirical evidence indicating that inflation is not only driven by energy prices but also by increasing mark-ups (Hayes & Jung 2022, Konczal & Lusiani 2022, Weber & Wasner 2023), where increases in mark-ups are in turn driven by the shift in consumer demand towards goodsproducing sectors and supply bottlenecks in these sectors. Third, periods of high inflation are periods of income redistribution. The struggle over who bears the cost of higher energy prices and excess demand is shaped by the relative pricing power of firms between sectors as well as the relative strength of labour and capital within sectors. Fourth, our policy analysis suggests that a corporate tax on windfall profits in the energy sector that are transferred to workers is effective in reducing inflation, while at the same time sustaining workers' real disposable income. By contrast, a policy of wage restraint reduces inflation but leads to a significant reduction in disposable income for the affected workers, on top of the fall in real wages due to higher energy prices. The reduction in real wages further amplifies the contractionary effects of the energy price hike on aggregate demand. A reduction in aggregate demand through contractionary policy does reduce inflation but amplifies the reduction in output and unemployment caused by the energy price hike. Put plainly: if a policy requires severe unemployment rates to bring down inflation, suspicions may abound regarding whether the disease is better than the cure.²⁸

Overall, our analysis points out that different anti-inflationary policies have different distributional consequences, which should be taken into account. In the short term, policies which reduce the conflict over the distribution of income, for example by channelling excess profits to workers hit by higher energy costs, can have a deflationary effect. In the medium term, addressing the root causes of energy-driven inflation seems to be the most promising approach. In this respect, our model highlights the key role of energy intensity as a factor that amplifies the adverse inflationary and distributional effects of shocks to global fossil fuel markets. Reducing energy intensity may thus not only help to decarbonise the economy but also stabilise inflation and income distribution in the face of global shocks. Our model furthermore highlights that increasing mark-ups can fuel inflation. Thus, policies that reduce the pricing power of firms should be considered. This is particularly important in the light of the recent increase in industrial action and wage demands in response to declining real wages. Limiting the pricing

 $^{^{28}}$ The Volcker shock has proven that high enough interest rates will eventually bring down inflation, but the negative side effects were substantial and indiscriminate. The US unemployment rate peaked at 10.8% in December 1982.

power of firms would dampen the fall in households' living standards and thereby reduce the risk of inflationary spirals. These conclusions are particularly important for countries like the US which has not introduced a windfall tax, but where the labour share of income has been declining (Guschanski & Onaran 2022) while income inequality has been increasing (Saez & Zucman 2020). They gain urgency in light of the rise in industrial action and wage demands in response to declining real wages after the energy price shock. The implications of our analysis contrast with a theoretical paradigm that is insensitive to the distributional drivers of inflation (Amiti et al. 2022, Barrett & Adams 2022, Hazell et al. 2022) and the distributional outcomes of anti-inflationary policies (Bolhuis et al. 2022).

There are several ways in which the framework presented here could be expanded in future research. Firstly, by explicitly modelling monetary policy one could compare an inflation targeting central bank with one which takes the distributional consequences into account. Secondly, a more direct treatment of inflation expectations could be added, e.g. by allowing for adaptive expectations along the lines of Hein & Stockhammer (2010) or Hein (2023). Thirdly, workers real wage targets could be endogenised to react to the state of the labour market. This requires a careful analysis of how quickly social norms around wages change and how sensitive wage demands are to the state of the economy. Fourth, the model could be extended by applying an agent-based household and firms sector (e.g. Caiani et al. 2016, Reissl 2020). This extension could be fruitful in order to analyse the heterogeneity in firm profitability at a more granular level, including, for example, potential changes in profit margins of firms providing luxury goods or consumer staples.

References

- Aboobaker, A. (2022), 'Macroeconomic determinants of south africa's post-apartheid income distribution', World Inequality Lab Working Papers (2022/08).
- Amiti, M., Heise, S., Karahan, F. & Sahin, A. (2022), 'Inflation strikes back: The return of wage to price pass-through', NBER Working Paper 31211. URL: https://www.nber.org/papers/w31211
- Argitis, G. & Dafermos, Y. (2011), 'Finance, inflation and employment: a postkeynesian/kaleckian analysis', *Cambridge Journal of Economics* 35(6), 1015–1033.
- Autor, D., Dorn, D., Katz, L. F., Patterson, C. & Reenen, J. V. (2020), 'The fall of the labor share and the rise of superstar firms*', *TheQuarterlyJournalofEconomics*135(2), 645--709.
- Bailey, A. (2022), 'think and reflect' before seeking pay rise', The Independent. URL: https://www.independent.co.uk/news/business/news/bank-of-england-pay-rise-andrewbailey-salary-b2080309.html
- Baqaee, D. & Farhi, E. (2022), 'Supply and demand in disaggregated keynesian economies with an application to the covid-19 crisis', *American Economic Review* **112**(5).
- Barkai, S. (2020), 'Declining labor and capital shares', *The Journal of Finance* **75**(5), 2421–2463.

URL: https://doi.org/10.1111/jofi.12909

- Barrett, P. & Adams, J. (2022), 'Shocks to inflation expectations', *IMF Working Paper No.* 22/72.
- Bernanke, B. & Blanchard, O. (2023), 'What caused the u.s pandemic-era inflation?', *Hutchins Center Working Paper 86*.
- Bhuller, M., Moene, K. O., Mogstad, M. & Vestad, O. L. (2022), 'Facts and fantasies about wage setting and collective bargaining', *Journal of Economic Perspectives* **36**(4), 29–52.
- Blanchard, O. (1986), 'The wage price spiral', *The Quarterly Journal of Economics* **101**(3), 543–566.
- Blecker, R. A. & Setterfield, M. (2019), Heterodox Macroeconomics, Edward Elgar.
- Bolhuis, M. A., Cramer, J. N. L. & Summers, L. H. (2022), 'Comparing past and present inflation', *NBER Working Paper 30116*.
- Bräuning, F., Fillat, J. L. & Joaquim, G. (2022), 'Cost-price relationships in a concentrated economy', Federal Reserve Bank of Boston Current Policy Perspectives. June 21, 2022.
- Burns, J., Quinn, M., Warren, L. & Oliveira, J. (2013), Management Accounting, McGraw-Hill Education - Europe.

- Caiani, A., Russo, A., Palestrini, A. & Gallegati, M., eds (2016), *Economics with Heterogeneous Interacting Agents*, Springer International Publishing.
- De Loecker, J., Eeckhout, J. & Unger, G. (2020), 'The rise of market power and the macroeconomic implications', The Quarterly Journal of Economics 135(2), 561–644. URL: https://academic.oup.com/qje/article/135/2/561/5714769
- di Giovanni, J., Kalemli-Özcan, S., Silva, A. & Yildirim, M. A. (2022), 'Quantifying the inflationary impact of fiscal stimulus under supply constraints', *NBER Working Paper 30892*
- Dogra, K., Heise, S., II, E. S. K., Meyer, B. H., Rich, R. W., Schoenle, R. S., Topa, G., van der Klaauw, W. & de Bruin, W. B. (2023), 'Estimates of cost-price passthrough from business survey data', *Federal Reserve Bank of New York Staff Reports* (1062).
- Domash, A. & Summers, L. (2022), 'The relation between nominal and real wage growth', Medium.com, April 11.

URL: https://medium.com/@alex.domash/the-relation-betweennominal-and-real-wagegrowth-2bfd2e1b27b8

Donovan, P. (2023), What is profit-led inflation?, Technical report, UBS Chief economist's comment (16 March 2023).

- Drury, C. (2017), Management and Cost Accounting, Cengage Learning.
- Drury, C. & Tayles, M. (2006), 'Profitability analysis in UK organizations: An exploratory study', The British Accounting Review 38(4), 405–425.
- Enache, C. (2022), 'What european countries are doing about windfall profit taxes', *Tax Foun*dation .
 - URL: https://taxfoundation.org/windfall-tax-europe/
- Flaschel, P. & Skott, P. (2006), 'Steindlian models of growth and stagnation', *Metroeconomica* **57**(3), 303–338.
- Franke, R. (2022), 'An empirical test of a fundamental harrod-kaldor business cycle model', Structural Change and Economic Dynamics 60, 1–14.
- Franke, R. & Westerhoff, F. (2012), 'Structural stochastic volatility in asset pricing dynamics: Estimation and model contest', *Journal of Economic Dynamics and Control* 36(8), 1193–1211.
- Furman, J. (2021), What the federal reserve should do now: An elaboration, in 'Paper prepared for delivery at the Peterson Institute for International Economics, November 18, 2021', Peterson Institute for International Economics.
- Guerrieri, V., Lorenzoni, G. & Straub, L. (2021), Monetary policy in times of structural reallo-

cation, *in* 'Prepared for the 2021 Jackson Hole Economic Policy Symposium', Federal Reserve Bank of Kansas City.

- Guschanski, A. & Onaran, O. (2022), 'The decline in the wage share: Falling bargaining power of labour or technologicalprogress? industry-level evidence from the oecd', *Socio-Economic Review* **20**(3), 1091–1124.
- Hamilton, J. D. (2018), 'Why you should never use the hodrick-prescott filter', The Review of Economics and Statistics 100(5), 831–843.
- Hayes, C. & Jung, C. (2022), 'Prices and profits after the pandemic', Institute for Public Policy Research and Common Wealth .
 URL: https://www.ippr.org/publications/prices-and-profits-after-the-pandemic
- Hazell, J., Herreño, J., Nakamura, E. & Steinsson, J. (2022), 'The slope of the phillips curve: Evidence from u.s. states', *The Quarterly Journal of Economics* **137**(3), 1299–1344.
- Hein, E. (2014), Distribution and Growth after Keynes: A Post-Keynesian Guide, Edward Elgar, Cheltenham Northampton, MA.
- Hein, E. (2023), Macroeconomics after Kalecki and Keynes, Edward Elgar Publishing.
- Hein, E. & Stockhammer, E. (2010), 'Macroeconomic policy mix, employment and inflation in a post-keynesian alternative to the new consensus model', *Review of Political Economy* 22(3), 317–354.
- Hunt, J. (2022), 'Chancellor warns unions not to jeopardise britain's recovery', Financial Times

URL: https://www.ft.com/content/251f11db-6fb0-4ef0-9678-bfec582589e9

IMF (2022), World economic outlook (october 2022), Technical report, International Monetary Fund.

URL: https://www.imf.org/en/Publications/WEO/Issues/2022/10/11/world-economic-outlook-october-2022

- Jorda, O., Liu, C., Nechio, F. & Rivera-Reyes, F. (2020), 'Why is u.s. inflation higher than in other countries?', Federal Reserve Bank of San Francisco Economic Letter 2022-07.
- Jorda, O. & Nechio, F. (2022), 'Inflation and wage growth since the pandemic', Federal Reserve Bank of San Francisco Working Paper 2022-17. URL: https://www.frbsf.org/economic-research/publications/working-papers/2022/17/
- Kaldor, N. (1985), *Economics without equilibrium*, M. E. Sharpe, Armonk, New York.
- Kilian, L. & Zhou, X. (2022a), 'The impact of rising oil prices on u.s. inflation and inflation expectations in 2020–23', *Energy Economics* 113.
- Kilian, L. & Zhou, X. (2022b), 'Oil prices, gasoline prices, and inflation expectations', Journal of Applied Econometrics 37(5), 867–881.

- Klenow, P. J. & Malin, B. A. (2010), 'Microeconomic evidence on price-setting', pp. 231–284.
- Konczal, M. & Lusiani, N. (2022), 'Prices, profits, and power: An analysis of 2021 firm-level markups', Roosevelt Institute Working Paper.
- Lavoie, M. (2014), Post-Keynesian Economics: New Foundations, Edward Elgar, Cheltenham, UK Northampton, MA.
- Lavoie, M. (2023), 'Some controversies in the causes of the post-pandemic inflation', medium.com
 - $\label{eq:urrel} \textbf{URL:} \quad https://medium.com/@monetarypolicyinstitute/some-controversies-in-the-causes-of-the-post-pandemic-inflation-1480a7a08eb7$
- Lorenzoni, G. & Werning, I. (2023), 'Inflation is conflict', NBER Working Paper 31099.
- Martins, G. K. & Skott, P. (2021), 'Sources of inflation and the effects of balanced budgets and inflation targeting in developing economies', *Industrial and Corporate Change* **30**(2), 409–444.
- Medina, J. P. & Soto, C. (2005), 'Oil shocks and monetary policy in an estimated dsge model for a small open economy', *Central Bank of Chile Working Papers* (353).
- Medio, A. & Lines, M. (2003), *Nonlinear Dynamics. A Primer*, Cambridge University Press, Cambridge, New York, Melbourne.
- Onaran, O. & Galanis, G. (2014), 'Income distribution and growth: A global model', Environment and Planning A: Economy and Space 46(10), 2489–2513.
- Ratner, D. & Sim, J. (2022), 'Who killed the phillips curve? a murder mystery', Finance and Economics Discussion Series 2022-028.
- Reissl, S. (2020), 'Minsky from the bottom up formalising the two-price model of investment in a simple agent-based framework', *Journal of Economic Behavior & Organization* **177**, 109– 142.
- Rowthorn, R. E. (1977), 'Conflict, inflation and money', *Cambridge Journal of Economics* 1, 215–239.
- Saez, E. & Zucman, G. (2020), 'The rise of income and wealth inequality in america: Evidence from distributional macroeconomic accounts', *Journal of Economic Perspectives* 34(4), 3–26.
- Sánchez, M. (2008), Oil shocks and endogenous markups. results from an estimated euro area dsge model, Technical Report 860, European Central Bank Working Paper Series.
- Setterfield, M. (2006), 'Balancing the macroeconomic books on the backs of workers: A simple analytical political economy model of contemporary u.s. capitalism', *International Journal of Political Economy* 35(3), 46–63.
- Setterfield, M. (2022), Inflation and distribution during the post-covid and recovery: a kaleckian and approach, Technical report, The New School for Social Research Department of Economics Working Paper 09/2022.

- Shapiro, A. H. (2022), 'Decomposing supply and demand driven inflation', Federal Reserve Bank of San Francisco Working Paper (2022-18). URL: https://doi.org/10.24148/wp2022-18
- Stachurski, J. & Sargent, T. J. (2022), Economic Networks: Theory and Computation, QuantEcon Book I. URL: https://networks.quantecon.org/
- Storm, S. (2022), 'Inflation in the time of corona and war', Institute for New Economic Thinking Working Paper No. 185.
- Szyld, D. B. (1985), 'Conditions for the existence of a balance growth solution for the leontief dynamic input-output model', *Econometrica* **53**(6), 1411–1420.
- Unite (2022), Corporate profiteering and the cost of living crisis, Unite the Union.
- Weber, I. M., Jauregui, J. L., Teixeira, L. & Pires, L. N. (2022), 'Inflation in times of overlapping emergencies: Systematically significant prices from an input-output perspective', University of Massachusetts Amherst Economics Department Working Paper Series.
- Weber, I. M. & Wasner, E. (2023), 'Sellers' inflation, profits and conflict: Why can large firms hike prices in an emergency?', **11**.

31

APPENDIX

A Corporate income taxation and lump sum transfers

Before analysing the effects of different policy interventions by means of numerical simulations, we formalise the introduction of a corporate income tax on profits in sector 1 (the energy sector) by applying a proportional tax on the sector's nominal profits ($\Pi_{1,n}$). Therefore nominal corporate income tax receipts (T_n) from sector 1 are given by:

$$T_n = t_1 \Pi_{1,n} \tag{A.1}$$

$$\Pi_{1,n} = p_1 Y_1 h_1 = p_1 Y_1 \left(1 - \frac{w_1}{v_1 p_1} \right), \tag{A.2}$$

where t_1 is the tax rate on sector 1 profits. The government uses these tax receipts and redistributes them as a lump sum payment to workers (for example through an energy bill support scheme). As a result, the consumption function and equilibrium nominal output become:

$$C_n = W_n + sT_n + (1 - s)\Pi_n.$$
(A.3)

$$Y_n^* = \frac{X_n}{s} \left(h - t_1 h_1 \epsilon \left[\delta_2 \gamma \mu^{\gamma - 1} + \delta_3 (1 - \gamma) \mu^{\gamma} \right] \right)^{-1}$$
(A.4)

Wage setting changes to:

$$w_{2,t+1} = w_{2,t} + w_{2,t} \left(\phi_2 \left[\omega_2^W - \omega_{2,t} - \alpha \frac{T_n}{p_t L_2} \right] \right)$$
(A.5)

$$w_{3,t+1} = w_{3,t} + w_{3,t} \left(\phi_3 \left[\omega_3^W - \omega_{3,t} - (1-\alpha) \frac{T_n}{p_t L_3} \right] \right), \tag{A.6}$$

where α is the share of tax receipts distributed to sector 2 workers $(1 - \alpha \text{ goes to sector 3 workers})$ and

$$\frac{T_n}{p_t L_i} = \frac{v_i T_n}{p_t Y_i}.$$
(A.7)

All other model equations remain the same.

В List of variables and parameters

 Table A1: Exogenous model parameters

Parameter	Description
X	autonomous components of demand
s	saving out of profits
δ_i	energy intensity of production
v_i	output per worker
θ_2^T	target mark-up in sector 2
ϵ	real price of energy
ω_1	real consumption wage in the energy sector
ω_i^T	workers' real wage target
ϕ_i	responsiveness of workers to deviations of real wage from target
ψ_i	responsiveness of firms to deviations of profit margin from target
γ	share of sector 2 in nominal final output

Table A2: List of model variables and education	equilibria (for the two-sector version)
---	---

Variable	Description	Equilibrium expressions
П	Real profit bill	$\Pi = \frac{X}{s}$
W	Real wage bill	$W = \frac{X}{s} \frac{1-h}{h}$
C	Real aggregate consumption	$C = \frac{X}{s} \frac{1-sh}{h}$
Y_1	Real intermediate good production	$Y_1 = \frac{\delta X}{sh}$
h	Profit share in aggregate	$h = 1 - \delta \frac{\omega_1}{v_1} - \frac{\omega_2}{v_2}$
h_1	Profit share in sector 1	$h_1 = 1 - \frac{\omega_1}{v_1 \epsilon}$
h_2	Profit share in sector 2	$h_2 = 1 - \frac{\omega_2}{v_2(1 - \delta\epsilon)}$
Y_2^*	Real final output in equilibrium	$Y_2^* = \frac{X}{sh}$
$\hat{p_2}$	Inflation rate in sector 2	$\hat{p}_2 = v_2 \psi_2 [r_2^T - (1 - \frac{\omega_2}{v_2} - \delta \epsilon)]$
r_1	Profit margin in sector 1	$r_1 = 1 - \frac{\omega_1}{\epsilon v_1} = h_1$
r_2	Profit margin in sector 2	$r_2 = 1 - \frac{\omega_2}{v_2} - \delta\epsilon$
r_2^T	Target profit margin in sector 2	$r_2^T = rac{ heta_2^T}{1+ heta_2^T}$
ω_2^*	Equilibrium real wage in sector 2	$\omega_2^* = \frac{\phi_2 \omega_2^{tW} + \psi_2 v_2 (1 - r_2^T - \delta \epsilon)}{\psi_2 + \phi_2}$

Variable	Description	Equilibrium expressions
Y^*	Real aggregate output in equilib-	$Y^* = \frac{X}{sh}$
	rium	
Y_1	Real intermediate good output	$Y_1 = [\gamma \delta_2(\mu)^{\gamma-1} + (1-\gamma)\delta_3(\mu)^{\gamma}]Y$
Y_2	Real output in sector 2	$Y_2 = \gamma(\mu)^{\gamma - 1} Y$
Y_3	Real output in sector 3	$Y_3 = (1 - \gamma)(\mu)^{\gamma} Y$
h	Profit share in aggregate	$h = 1 - (1 - \gamma)(\mu)^{\gamma}(\frac{\omega_3}{v_3}) - \gamma(\mu)^{\gamma-1}\frac{\omega_2}{v_2} -$
		$[\gamma \delta_2(\mu)^{\gamma-1} + (1-\gamma)\delta_3(\mu)^{\gamma}](\frac{\omega_1}{v_1})$
h_1	Profit share in sector 1	$h_1 = 1 - \left(\frac{\omega_1}{\epsilon v_1}\right)$
h_2	Profit share in sector 2	$h_2 = 1 - \left(\frac{\omega_2}{v_2((\mu)^{1-\gamma} - \epsilon \delta_2)}\right)$
h_3	Profit share in sector 3	$h_3 = 1 - \left(\frac{\omega_3}{v_3((\mu)^{-\gamma} - \epsilon \delta_3)}\right)$
r_1	Profit margin in sector 1	$r_1 = h_1$
r_2	Profit margin in sector 2	$r_2 = 1 - (\mu)^{\gamma - 1} \frac{\omega_2}{v_2} - (\mu)^{\gamma - 1} \delta_2 \epsilon$
r_3	Profit margin in sector 3	$r_3 = 1 - (\mu)^{\gamma} \frac{\omega_3}{v_3} - (\mu)^{\gamma} \delta_3 \epsilon$

Table A3: List of model variables and equilibria (for the three-sector version)

C Proof of proposition 2

The Jacobian matrix of the system in (17)-(18) is given by

$$J = \begin{bmatrix} \frac{\partial w_{2,t+1}}{\partial w_{2,t}} & \frac{\partial w_{2,t+1}}{\partial p_{2,t}} \\ \frac{\partial p_{2,t+1}}{\partial w_{2,t}} & \frac{\partial p_{2,t+1}}{\partial p_{2,t}} \end{bmatrix} = \begin{bmatrix} 1 + \hat{w}_2 - \phi_2 \omega_2 & \phi_2 \omega_2^2 \\ \psi_2 & 1 + \hat{p}_2 - \psi_2 \omega_2 \end{bmatrix}$$
(C.1)

To prove proposition 2, we need to show two things: firstly, that the system in (17)–(18) is unstable so that prices and nominal wages grow exponentially; and secondly, that prices and nominal wages grow at the same rate, i.e. that there is balanced growth. Mathematically, the first requires the Jacobian matrix to have a dominant eigenvalue that is outside the unit circle, and the second requires all elements of the dominant eigenvector associated with that eigenvalue to be nonzero and of the same sign. In that case, the dominant unstable eigenvalue will drive the dynamics of all state variables of the system such that these variables grow at the same rate, yielding constant and positive ratios between the state variables.

First, we analyse the conditions under which the Jacobian (C.1) is likely to have a dominant eigenvector with nonzero elements of the same sign. If the Jacobian matrix is nonnegative $(J \ge 0)$, a sufficient condition for balanced growth is that it is irreducible (Szyld 1985). In this case, by the Perron-Frobenius theorem, the dominant eigenvector of the Jacobian will be everywhere positive (Stachurski & Sargent 2022, pp.14-16). However, the Jacobian (C.1) will not necessarily be nonnegative as the elements J_{11} and J_{22} may become negative. Thus, irreducibility alone will not strictly guarantee balanced growth. Still, the occurrence of balanced growth is highly likely if the Jacobian matrix is irreducible even if some elements are negative, because then all state variables will feed into each other. Another way of looking at this is that if the Jacobian matrix is irreducible, the associated directed graph of the matrix is strongly connected (Szyld 1985, p. 1415), i.e. any node that represents an endogenous variable of the system can be reached from any other (see the nodes for w_2 and p_2 in Figure 2). As all variables feed into each other, they are likely to exhibit the same growth rate.

In the two-dimensional case, irreducibility requires the Jacobian to be non-triangular. Nontriangularity requires the off-diagonal elements of the Jacobian (C.1) to be non-zero. With an economically meaningful, i.e. positive, solution for the equilibrium real wage, non-triangularity is satisfied. Using the equilibrium solution for the real wage (23), firms' implicit real wage target $\omega_2^F = v_2 \left(\frac{1}{1+\theta_2} - \epsilon \delta\right)$, and imposing $\omega_2^* > 0$ yields the following condition:

$$\phi_2 \omega_2^W + \psi_2 \omega_2^F > 0. \tag{C.2}$$

Second, we derive the conditions under which the system in (17)–(18) is unstable. A necessary condition for two-dimensional discrete time dynamic systems to be stable is 1-tr(J)+det(J) > 0, where tr(J) and det(J) are the trace and determinant of the Jacobian matrix (Medio & Lines 2003, p. 52). Thus, if 1 - tr(J) + det(J) < 0, the system in (17)–(18) will be unstable. With $\hat{p}_2 = \hat{w}_2$, we have:

$$tr(J) = 2 + 2\hat{w}_2 - \omega_2(\phi_2 + \psi_2)$$
$$det(J) = 1 - \omega_2(\psi_2 + \phi_2) + \hat{w}_2[\hat{w}_2 + 2 - \omega_2(\psi_2 + \phi_2)].$$

From this, we get:

$$1 - tr(J) + det(J) = \hat{w}_2[\hat{w}_2 - \omega_2(\psi_2 + \phi_2)].$$
(C.3)

Using the equation for wage dynamics (17) and the equilibrium solution for the real wage (23) and simplifying yields:

$$1 - tr(J) + det(J) = -\hat{w}_2 \left[\frac{\omega_2^W \phi_2^2 + (2\phi_2 + \psi_2)\psi_2 \omega_2^F}{\phi_2 + \psi_2} \right].$$
 (C.4)

With a positive equilibrium inflation rate $\hat{p}_2 = \hat{w}_2 > 0$, this term is always negative, and the system thus unstable. By (24), the equilibrium inflation rate is positive if $\omega_2^W > \omega_2^F$, which proves proposition 2.

D Analytical discussion of the three-sector model

The Jacobian for the system in (46) - (49) is given by

$$\mathbf{J}(\mathbf{w}_{2}, p_{2}, w_{3}, p_{3}) = \begin{bmatrix} 1 + \hat{w}_{2} - \phi_{2}\omega_{2} & \frac{\gamma\phi_{2}w_{2}^{2}p_{3}^{\gamma-1}}{p_{2}^{\gamma+1}} & 0 & (1 - \gamma)\phi_{2}w_{2}^{2} \left(\frac{p_{3}^{\gamma}}{p_{2}^{\gamma}}\right) \\ \psi_{2} & 1 - \psi_{2}v_{2}(1 - r_{2}^{T}) + \psi_{2}v_{2}\epsilon_{1}\gamma\delta_{2} \left(\frac{p_{2}}{p_{3}}\right)^{\gamma-1} & 0 & \psi_{2}v_{2}\epsilon_{1}(1 - \gamma)\delta_{2} \left(\frac{p_{2}}{p_{3}}\right)^{\gamma} \\ 0 & \gamma\phi_{3}w_{3}^{2}\frac{p_{3}^{\gamma-1}}{p_{2}^{\gamma+1}} & 1 + \hat{w}_{3} - \phi_{3}\omega_{3} & (1 - \gamma)\phi_{3}w_{3}^{2}\frac{p_{3}^{\gamma-2}}{p_{2}^{\gamma}} \\ J_{41} & J_{42} & J_{43} & J_{44} \end{bmatrix}$$

$$(\mathbf{D}.1)$$

with

$$J_{41} = \frac{\partial p_{3,t+1}}{\partial w_2} = -\frac{ac\frac{\partial h}{\partial w_2}}{(bh+c)^2} = \frac{ac\gamma}{p_2 v_2 (bh+c)^2} \tag{D.2}$$

$$J_{42} = \frac{\partial p_{3,t+1}}{\partial p_2} = d\gamma \left(\frac{p_2}{p_3}\right)^{\gamma-1} - \frac{af\frac{\partial h}{\partial p_2} - ah\frac{\partial f}{\partial p_2}}{f^2} \tag{D.3}$$

$$J_{43} = \frac{\partial p_{3,t+1}}{\partial w_3} = \psi_3 - \frac{ac\frac{\partial h}{\partial w_3}}{(bh+c)^2} = \psi_3 + \frac{ac(1-\gamma)}{p_3 v_3 (bh+c)^2}$$
(D.4)

$$J_{44} = \frac{\partial p_{3,t+1}}{\partial p_3} = 1 + d(1-\gamma) \left(\frac{p_2}{p_3}\right)^{\gamma} - \frac{a_2 f_2 \frac{\partial h}{\partial p_3} - a_2 \tilde{h} \frac{\partial f_2}{\partial p_3}}{f_2^2} \tag{D.5}$$

where: $a = v_3 \psi_3 Y_3^P s p_{3,t}; a_2 = v_3 \psi_3 Y_3^P s; b = (1 + \beta_0) Y_3^P s; c = \beta_1 (1 - \gamma) X p_{2,t}^{\gamma} p_{3,t}^{-\gamma}; c_2 = \beta_1 (1 - \gamma) X p_{3,t}^{\gamma}; c_3 = \beta_1 (1 - \gamma) X p_{2,t}^{\gamma}; d = \psi_3 v_3 \epsilon_1 \delta_3; f = bh^* + c_2 p_{2,t}^{\gamma}; f_2 = bh^* + c_3 p_{3,t}^{-\gamma}; \tilde{h} = p_{3,t} h^*; \frac{\partial f}{\partial p_2} = b \frac{\partial h^*}{\partial p_2} + c_2 \gamma p_{2,t}^{\gamma-1}; \frac{\partial f_2}{\partial p_3} = b \frac{\partial h^*}{\partial p_3} - c_3 \gamma p_{3,t}^{-\gamma-1}; \frac{\partial \tilde{h}}{\partial p_3} = h^* + p_3 \frac{\partial h^*}{\partial p_3}.$

As discussed in Appendix C, for the system to be unstable, the Jacobian matrix needs to exhibit at least one root outside the unit circle. In the two-sector benchmark, instability is the outcome of the interaction between p_2 and w_2 , with workers and firms pursing inconsistent real wage targets. In the three-sector extension, this is not necessarily the case anymore as workers target the consumption real wage, whereas firms target the product real wage. However, the interaction between the two subsystems (p_2, w_2) and (p_3, w_3) introduces another potential source of instability. While it is difficult to derive economically meaningful instability conditions, numerical analysis based on the empirically calibrated parameters reported in section 3 shows that the two subsystems are individually stable while the combined system is unstable.

To show that the state variables are likely to grow at the same rate, we examine whether the Jacobian matrix D.1 is irreducible. As discussed in Appendix C, irreducibility makes it highly likely that the dominant eigenvector will have nonzero elements of the same sign (although with the presence of negative elements in the Jacobian, there is no guarantee that this will be the case). The directed graph associated with the Jacobian matrix (D.1) is shown in Figure A1. It can be seen that the directed graph is strongly connected, i.e. any node can be reached from any other, which means that (D.1) is irreducible (Szyld 1985, p. 1415). Indeed, the two subsystems (p_2, w_2) and (p_3, w_3) mutually impact each other. As a result, the elements of the dominant eigenvector are likely to be different from zero. If they have the same sign, all variables will grow at the same rate. For the empirically calibrated parameterisation, this is indeed the case.





Notes: Arrows from the state variables to themselves representing the main diagonal of the matrix (D.1) are omitted for clarity.

E Equilibria of the three-sector model

Equating the growth rates of wages and prices for each sector in (46)-(48) and (47)-(49) yields:

$$\omega_2 = \frac{c_1 - c_2 \mu^{1-\gamma}}{\phi_2 + \psi_2 \mu^{1-\gamma}} \tag{E.1}$$

$$\omega_3 = \frac{c_8 - c_9 \mu^{-\gamma} + \frac{c_{10}}{c_{11} + \frac{1}{\mu^{\gamma} + c_4 - c_5 \omega_3 - c_6 \omega_2 \mu - c_7 \mu}}}{\psi_3 \mu^{-\gamma} + \phi_3},$$
(E.2)

and equating the growth rates of wages in sector 2 and 3 in (46) and (47) yields:

$$\omega_2 = c_{12} + c_{13}\omega_3, \tag{E.3}$$

where the c_i 's are the following exogenous composite parameters:

$$c_{1} = \phi_{2}\omega_{2}^{W} + \psi_{2}v_{2}(1 - r_{2}^{T})$$

$$c_{2} = \phi_{2}v_{2}\delta_{2}\epsilon$$

$$c_{3} = \frac{\beta_{1}(1 - \gamma)X}{Y_{3}^{P}s}$$

$$c_{4} = -\delta_{3}(1 - \gamma)\left(\frac{\omega_{1}}{v_{1}}\right)$$

$$c_{5} = \frac{1 - \gamma}{v_{3}}$$

$$c_{6} = \frac{\gamma}{v_{2}}$$

$$c_{7} = \delta_{2}\gamma\left(\frac{\omega_{1}}{v_{1}}\right)$$

$$c_{8} = \phi_{3}\omega_{3}^{W}$$

$$c_{9} = \epsilon\delta_{3}\psi_{3}v_{3}$$

$$c_{10} = \psi_{3}v_{3}$$

$$c_{11} = 1 + \beta_{0}$$

$$c_{12} = \omega_{2}^{W} - \frac{\phi_{3}}{\phi_{2}}\omega_{3}^{W}$$

$$c_{13} = \frac{\phi_{3}}{\phi_{2}}$$

Combining (E.1) and (E.3) yields:

$$\omega_3 = \frac{c_1 - c_2 \mu^{1-\gamma}}{(\phi_2 + \psi_2 \mu^{1-\gamma})c_{13}} - \frac{c_{12}}{c_{13}}.$$
(E.4)

Combining (E.2), (E.3), and (E.4) yields:

$$\omega_{3} = \frac{c_{8} - c_{9}\mu^{-\gamma} + \frac{c_{10}}{c_{11} + \frac{c_{10}}{\mu^{\gamma} + c_{4} - \left(\frac{c_{1} - c_{2}\mu^{1-\gamma}}{\phi_{2} + \psi_{2}\mu^{1-\gamma}}\right)\left(c_{6}\mu + \frac{c_{5}}{c_{13}}\right) + \frac{c_{5}c_{12}}{c_{13}} - c_{7}\mu}}{\psi_{3}\mu^{-\gamma} + \phi_{3}}.$$
 (E.5)

Equations (E.4) and (E.5) are a two-dimensional nonlinear system in ω_3 and μ . While this can be reduced to a single rational function in μ whose roots yield the equilibria, the resulting expression does not admit a clear-cut statement about the possible number of economically meaningful equilbria. Instead, Figure A2 plots equations (E.4) and (E.5) for the empirically calibrated parameters reported in Table 1. It can be seen that there are three equilibria. The first two equilibria occur for very small values of μ and values of ω_3 above the real wage target ω_3^W . Thus, these equilibria are inconsistent with a positive equilibrium inflation rate. The economically meaningful equilibrium occurs for $\omega_3^* < \omega_3^W$ and is the one reached in the simulations reported in the main text.

Figure A2: Equilibria of ω_3 and μ



Notes: Based on empirically calibrated parameters reported in Table 1. Dashed line: equation (E.4), solid line: equation (E.5), vertical dotted line: economically meaningful equilibrium attained in simulation.

F Data for calibration

The energy sector consists of mining (NAICS 21), which is dominated by oil and natural gas extraction. The services sector includes utilities (22), information (51), finance and insurance (52), real estate (53), professional services (54), management of companies (55), administrative and support services (56), education (61), health care (62), arts and entertainment (71), accommodation (72) and other services $(81)^{29}$. This represents a narrowly defined services sector which does not include what we call 'goods related' services such as transport or retail. Since these latter sectors would also benefit from a demand shift we included them in a broadly defined goods sector consisting of agriculture (11), construction (23), manufacturing (31-33), wholesale trade (41-42), retail trade (44-45) and transportation and warehousing (48-49).

Series	Definition	BEA Table	Notes
Real aggregate output	Log of real gross output, detrended	Value added by industry	Detrended using Hamilton's regression filter
Sector inflation	Annual growth rate of implicit sector deflator	Value added by industry	
Aggregate price index	Deflator of S2 and S3 output	Value added by industry	
Sector profit shares	Corporate profits relative to value added	Table 6.17D	
Real energy price index	Price index of S1 over aggregate price index	Value added by industry	Energy price index was normalised to have mean 0.7 and standard deviation 0.01.

All series except the serieson corporate profits are obtained from BEA's GDP by Industry Tables.

 $^{^{29}\}mathrm{Numbers}$ in brackets represent the NAICS 2 digit code. Some sector names are abbreviated.



Figure A3: Time series used to construct empirical moments

Notes: S1: sector 1 (energy), S2: sector 2 (services), S3: sector 3 (goods).

- This paper builds and calibrates a novel three-sector conflict inflation model
- In our model periods of high inflation lead to significant redistribution of income
- Energy shocks drive increases in energy sector's profit margins
- Non-energy profits are driven by COVID-induced demand shifts and supply bottlenecks
- Taxing windfall profits to stabilize workers' disposable income reduces inflation

Conceptualization: all

Methodology: all

Software: KK and RW

Data Curation: all

Writing - Original Draft: all

Writing – Review & Editing: all